Author’s response to reviews

Title: Targeted therapy in pulmonary veno-occlusive disease: time for a rethink?

Authors:
Qin Luo (luoqin2009@163.com)
Qi Jin (drjinqi@foxmail.com)
Zhihui Zhao (1250167892@qq.com)
Qing Zhao (zhaoqingfw@aliyun.com)
Xue Yu (yuxuefw@foxmail.com)
Lu Yan (Ripple_Yan@163.com)
Yi Zhang (med-zhangyi@foxmail.com)
Changming Xiong (xiongcm2000@163.com)
Zhihong Liu (zhihongliufuwai@163.com)

Version: 3 Date: 08 Dec 2019

Author’s response to reviews:

Dear Editor,

Many thanks for your efforts in reviewing our revised manuscript and providing helpful suggestions. Our detailed point-by-point responses to the comments are as follows:

1. -- In order to help protect participant anonymity, please limit the identifiable information presented in Table S2. A good way to do this is to present the age of the participant as a range (e.g. 20's-30's) or removing the age (or sex) information.
   We ask this, as it is BioMed Central policy to not publish more than 2 indirect identifiers (age, sex, profession, etc.) without explicit consent for publication from the participants.
   Currently, the age and gender information in conjunction with other details, may compromise patient anonymity.
Authors: Thanks for your good advice, we removed the age information in Table S2 to avoid any compromise in patient anonymity.

2. -- In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.
Authors: Thank you for your comments. We have added it in the Funding section by stating that “The funders had no role in the design of the study, the collection, analysis, and interpretation of data and in writing the manuscript.”
3. -- Thank you for providing a response to the reviewers. As this document is no longer required at this stage of the publication process, please remove it from your submission’s supplementary files. Authors: Thanks for your advice, this document was removed from supplementary files.

4. -- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Authors: A single, final, clean version of our manuscript was uploaded.

Thanks once again for carefully reviewing our paper. We hope that the revised version is acceptable.

Best Regards
Qi Jin, on the behalf of the authors. Center for Pulmonary Vascular Diseases, Fuwai Hospital, National Center for Cardiovascular Diseases, Chinese Academy of Medical Sciences and Peking Union Medical College, 167 Beilishi Road, Xicheng District, Beijing, 100037, China. E-mail: drjinqi@foxmail.com.